////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-AttributedCharacterIterator]]++AttributedCharacterIterator++::
* ++[[painless-api-reference-AttributedCharacterIterator-getAllAttributeKeys-0]]<<painless-api-reference-Set,Set>> link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getAllAttributeKeys%2D%2D[getAllAttributeKeys]()++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getAllAttributeKeys%2D%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getAttribute-1]]def link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getAttribute%2Djava.text.AttributedCharacterIterator$Attribute%2D[getAttribute](<<painless-api-reference-AttributedCharacterIterator-Attribute,AttributedCharacterIterator.Attribute>>)++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getAttribute%2Djava.text.AttributedCharacterIterator$Attribute%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getAttributes-0]]<<painless-api-reference-Map,Map>> link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getAttributes%2D%2D[getAttributes]()++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getAttributes%2D%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getRunLimit-0]]int link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getRunLimit%2D%2D[getRunLimit]()++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getRunLimit%2D%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getRunLimit-1]]int link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getRunLimit%2Djava.util.Set%2D[getRunLimit](<<painless-api-reference-Set,Set>>)++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getRunLimit%2Djava.util.Set%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getRunStart-0]]int link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getRunStart%2D%2D[getRunStart]()++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getRunStart%2D%2D[java 9])
* ++[[painless-api-reference-AttributedCharacterIterator-getRunStart-1]]int link:{java8-javadoc}/java/text/AttributedCharacterIterator.html#getRunStart%2Djava.util.Set%2D[getRunStart](<<painless-api-reference-Set,Set>>)++ (link:{java9-javadoc}/java/text/AttributedCharacterIterator.html#getRunStart%2Djava.util.Set%2D[java 9])
* Inherits methods from ++<<painless-api-reference-CharacterIterator,CharacterIterator>>++, ++<<painless-api-reference-Object,Object>>++
